Author's response to reviews

Title: Infective Endocarditis caused by staphylococcal endocarditis in a patient with Atopic dermatitis-a case report and literature review

Authors:

Gadha Mohiyiddeen (gaadha@yahoo.com)
Ian Brett (ian.brett@tgh.nhs.uk)
Edward Jude (Edward.Jude@tgh.nhs.uk)

Version: 7 Date: 16 December 2007

Author's response to reviews:

Few final editing has been made with respect to language, format and typography. Abstract is modified with subtitles - introduction, case presentation and conclusion. A brief introduction is added before the case presentation. A comment about patients consent is included after authors contribution and figure legends after the list of references.